Author’s response to reviews

Title: Chronic disseminated candidiasis manifesting as hepatospelnic abscesses among patients with hematological malignancies

Authors:

Chien-Yuan Chen (chienyuanchen@ntu.edu.tw)
Aristine Cheng (arischeng@ntuh.gov.tw)
Feng-Ming Tien (b92401007@gmail.com)
Po-Chu Lee (pchee@ntuh.gov.tw)
Hwei-Fang Tien (hftien@ntu.edu.tw)
Wang-Huei Sheng (whsheng@ntu.edu.tw)
Yee-Chun Chen (yeechunchen@ntuh.gov.tw)

Version: 3 Date: 17 Apr 2019

Author’s response to reviews:

INFD-D-18-00434R2

Chronic disseminated candidiasis manifesting as hepatospelnic abscesses among patients with hematological malignancies

Chien-Yuan Chen; Aristine Cheng; Feng-Ming Tien; Po-Chu Lee; Hwei-Fang Tien; Wang-Huei Sheng; Yee-Chun Chen

BMC Infectious Diseases

Dear Editor:

1. Before we send this back to the original reviewer for re-review, we recommend that you check copy-editing with a native English speaker.

Reply: We had revised with a native English speaker.

2. Please make sure that where possible, you do not only provide a response to the concerns raised, but also implement required changes in your manuscript, as suggested by the reviewer.
Reply: Thank you, we had made the major revision. The cause of microabscess resolution is related to recovery of neutropenia, not related to neutropenia.

3. We have detected a significant level of text overlap between your manuscript and the published article Chien-Yuan Chen, et al (2018) “Clinical characteristics and treatment outcomes of pulmonary invasive fungal infection among adult patients with hematological malignancy in a medical centre in Taiwan, 2008–2013”. Journal of Microbiology, Immunology and Infection. This level of overlap is a breach of the COPE guidelines BMC adheres to. Please reword the overlapping text in your revised manuscript.

Whilst I appreciate that some of the text overlap was in the methods and replicated from previous studies performed by the authors of this paper, I ask that you please make sure to keep the overlap to a minimum. In the methods section, please always state that the methods have already been described elsewhere and provide a citation.

Reply: We add the description and reference citation in the methods section.

4. If no funding was provided for this study, please write “not applicable” in the funding section.

Reply: We revised it.

Sincerely

Chien-Yuan Chen  Md, PhD